Author's response to reviews

Title: Plasma leptin and insulin-like growth factor I levels during acute exacerbations of Chronic Obstructive Pulmonary Disease

Authors:

Prokops Kythreotis (pkc@otenet.gr)
Aggeliki Kokkini (pkc@otenet.gr)
Stavrina Avgeropoulou (stavrinavg@yahoo.gr)
Argyro Hadjioannou (arghadjioanou@yahoo.gr)
Efgenia Anastasakou (efganastasakou@yahoo.gr)
Antonis Rasidakis (rasid@ath.forthnet.gr)
Petros Bakakos (petros44@hotmail.com)

Version: 5 Date: 9 February 2009

Author's response to reviews: see over
Dear editor,

Following your notification on acceptance of our manuscript entitled “Plasma leptin and insulin-like growth factor I levels during acute exacerbations of Chronic Obstructive Pulmonary Disease”, we are pleased to resubmit the final version including all the formatting changes required.

1. The full affiliations have been added.
2. There are no acknowledgements, medical writer or funding sources.
3. Reference 3 has been corrected
4. Tables have been checked.
5. Minor revisions have also been made according to request.

Yours sincerely,

Petros Bakakos
Lecturer in Respiratory Medicine
Athens University